

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Franco, Chiara; Weche Gelubcke, John P.

Working Paper The death of German firms: What role for foreign direct investment?

Working Paper Series in Economics, No. 264

Provided in Cooperation with: Institute of Economics, Leuphana Universität Lüneburg

Suggested Citation: Franco, Chiara; Weche Gelubcke, John P. (2013) : The death of German firms: What role for foreign direct investment?, Working Paper Series in Economics, No. 264, Leuphana Universität Lüneburg, Institut für Volkswirtschaftslehre, Lüneburg

This Version is available at: https://hdl.handle.net/10419/73154

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Ш Д 4 1 VORKING

The death of German firms: What role for foreign direct investment?

by

Chiara Franco and John P. Weche Gelübcke

University of Lüneburg Working Paper Series in Economics

No. 264

February 2013

www.leuphana.de/institute/ivwl/publikationen/working-papers.html

ISSN 1860 - 5508

The death of German firms: What role for foreign direct investment?*

Chiara Franco[†] John P. Weche Gelübcke[‡]

February 2013

Abstract

This paper aims at examining the role played by inward Foreign Direct Investments (FDI) in affecting the exit probabilities of German manufacturing firms in the precrisis year 2007. We introduce two main novelties: in the first place, we include the FDI variable, dividing it between types of foreign investor (industrial vs. financial) besides the usual analysis with the division by country of origin. Secondly, we analyze whether FDI may have effects not only on the probability that a firm exits the domestic market, but also on whether it stops being internationally involved, that is, whether it stops importing or exporting. We find that German firms in most cases suffer from higher competition introduced by foreign firms except when they are part of a high-R&D region or a high-tech sector when they have the needed absorptive capacity to take advantage of possible spillover effects. We also find that U.S. FDI has a crowding out effect for firms located in low-tech sectors but not in high-tech sectors. The results are reversed when considering financial investments instead of industrial investments. Finally, we find that FDI is negatively correlated with exits from export markets but positively with those from import markets.

JEL Classification: F23; L60; O30.

Keywords: MNE, FDI, foreign ownership, survival, Germany.

 $^{^*}$ We would like to thank participants of the 7th workshop of the International Study Group on Exports and Trade (ISGEP) 2012 in Stockholm for useful comments and suggestions. Usual caveats apply.

 $^{^\}dagger\,$ Catholic University of the Sacred Heart, Milan, Italy, E-mail:chiara.franco@unicatt.it.

[‡] Leuphana University Lüneburg, Germany, E-mail: wechegel@uni.leuphana.de.

1 Introduction

The increasing pace of globalization is putting progressively more emphasis on the role played by trade and foreign direct investments (FDI) in affecting market dynamics. After the initial empirical contributions, among which the most relevant was Mansfield (1962), who found a negative relation between the initial size and the growth of a firm¹, the industrial organization literature has started, from the 1980s, to focus on the study of the determinants of firm survival rates. Most of these studies have used both firm and industry level determinants to explain the exit probability of firms from the market. Among the former, we usually find age, size, total factor productivity (TFP) or labour productivity, R&D investment, and firm ownership.² Among the industry determinants, the variables that are usually included are market concentration or technological intensity, whereas empirical research has not devoted much analysis to how the sectoral presence of foreign firms can affect the survival of local firms. Indeed, when multinational enterprises (MNEs) choose FDI as a mode of entry into the host market, the superior firm-specific assets owned by those firms could spill over to domestic firms and therefore increasing their performance. This theoretical expectation has led many governments to adopt policies to stimulate FDI inflows. The majority of empirical analyses have concentrated their efforts on the investigation of the likely indirect impacts of FDI on local firms, making use of the concept of productivity spillovers: that is, by measuring whether an industry level measure of FDI can affect TFP of local firms. However, looking at the large body of empirical evidence gathered so far, only mixed results have been found, regardless of the country considered. Most of the time it has not been considered that foreign firms can also reshape the market forces causing a higher competition effect to emerge, a consequent shrinkage of profit margins, and, in the end, their own exit

¹ This result was against Gibrat's Law, stated in 1931, according to which the size of a firm is irrelevant in explaining its performance.

² The literature examining the issue of firm survival first aimed at comparing the dynamics of the survival of foreign and local firms to investigate whether the two dynamics could be different in some way. The general finding is that foreign MNEs are more likely to survive than domestic market oriented firms, as they are bigger, more R&D intensive, and more productive.

from the market. Two main mechanisms are at work: the first implies an increase in the efficiency of the sector, causing the least efficient domestic firms to leave the market; the second is the transmission of a spillover effect to domestic firms, causing an increase in their performance and therefore allowing them to stay in the market despite the greater competition. Which of the two effects is going to dominate is an empirical matter. Only a few papers have investigated this research question. For example, Görg and Strobl (2003b), examining the case of Ireland, found that only in high-tech sectors does the presence of foreign firms not cause local firms to exit from the market.

Following this branch of the literature, which goes beyond the usual debate about an FDI productivity spillover, we analyze whether FDI may influence the exit probability of domestic firms in Germany. In doing this, we make use of a tailor made dataset of manufacturing enterprises in which we are able to distinguish between foreign and domestic firms. The present paper contributes to the literature in several ways: first, it provides the first empirical evidence of the FDI survival effect for the German economy. We choose to focus on the German case mainly because this European country is one of the leading destinations, worldwide, for FDI, and the share of employees working for foreign firms is remarkably high: more than 10%(Nahm (2011)). The way the dataset is built allows us to analyze exits in 2007, giving us the opportunity to test how the presence of foreign firms shaped the market before the crisis started. The crisis period has recently been under scrutiny: for example, Godard et al. (2012) analyzed the behavior after 2008 of foreign and local firms in the Irish case. Similarly, Alfaro and Chen (2012) examined how foreign affiliates around the world responded to the crisis in comparison to local firms, finding that foreign firms had more chances of survival than local counterfactuals with similar economic characteristics. However, the time period just immediately before the crisis has not been explored by the FDI survival literature.

The second main contribution is that we introduce heterogeneity both on the part of the domestic firms, accounting for their international status (exporters, importers, and two-way traders) and on the part of the MNEs, that is, by considering FDI from different countries and considering different types of foreign ownership (industrial vs. financial). Third, we do not limit our analysis to the effect of FDI on exit from the domestic market, but also study whether firms stop being involved in exporting or importing activities, that is, whether they lose their international status. This research question has been rarely examined: for example, Alvarez and Lopez (2008) analyze the determinants of the exit from export and import markets in the Chilean case but without considering the role played by FDI.

Although we only have cross-sectional data, and therefore we can interpret our results only in terms of correlations, our main findings point to a negative effect of FDI on the survival of domestic firms: the effect is the same when we split the sample between East and West Germany and is also independent of the international status of the firm. However, the size of the coefficient for West Germany indicates that the displacement effect is higher in this part of the country than in East Germany. Local firms are more likely to survive when they belong to a high-tech sectors and a high-tech region: this is in accordance with the fact that firms that are part of those categories benefit from high spillover effects, being in the position of balancing the clearing effect due to their higher absorptive capacities. It is also interesting to note that when we combine FDI heterogeneity with local firm heterogeneity, we find that industrial FDI generates a crowding out effect only in low-tech industries, unlike financial FDI, which mainly generates positive spillover effects. When coming to consider the effect of FDI on the probability of firms' pursuing their international activities, we find that exporting firms more easily continue being exporters while importing firms can be forced to give up this activity, even though the effect seems to be driven by firms in East Germany.

The remainder of this paper is organized as follows: Section 2 reviews the literature about the role played by FDI in affecting local firms' survival; Section 3 describes the dataset; Section 4 presents the econometric methodology and covariates used. Section 5 reports the probit estimates, and Section 6 offers some conclusive remarks.

2 FDI and firm survival: The related literature

The literature concerning the impact of FDI on local firms started more than a decade ago, with the aim of exploring the effects of foreign-owned firms on the productivity of local firms (e.g., Aitken and Harrison (1999)). The empirical approach followed requires the estimation of a production function augmented with a sectoral measure of FDI. The empirical analyses produced through this approach have mainly achieved mixed results, especially when taking into consideration the horizontal spillover effect rather than the vertical effect, for which positive results are usually found (e.g., Javorcik (2004)).³ Therefore, the efforts to investigate this multifold concept are still under way. However, the empirical approach used, which implies the measurement of the spillover effects in terms of TFP, can be quite narrow: therefore, some related strands of the literature have started to explore different methodologies to go beyond this approach. In the first place, the export spillover effect, that is, the effect played by MNEs on the exporting activities of the local firms (e.g., Kneller and Pisu (2007)), has been one of the first to be examined. The main research question revolves around the possibility that foreign firms influence both the export decisions and the export intensities of local firms. In a similar way, the second empirical approach that has recently emerged has concentrated attention on the explanation of the innovation spillover effect coming from FDI (e.g., Garcia et al. (2012)). In this case, the dependent variable is not the amount of exports, but rather the innovation activities of the firms, such as the product innovation counts or the number of patent applications. The last strand of the literature trying to explore further avenues of research is related to the investigation of the effect of MNEs on the probability of local firms' survival. Nevertheless, a similar but distinct research question was first examined: at first, authors concentrated on explaining what are the effects of foreign ownership on the probability of exit of the same firm

³ In the case of the horizontal measure, the FDI variable is represented as the share of foreign firm sales or employees in total sales or in total employees of the sector under consideration. Instead, in the vertical FDI measure, the variable is built by multiplying the horizontal measure by the amount of production supplied or sold to the upstream, respectively, downstream sector.

from the market. In this respect, two opposing views emerged: according to the first, MNEs should display higher probabilities of survival in the market, as they are more likely to invest in more technology intensive sectors, they are bigger, and they are more productive. Moreover, the role played by sunk costs is quite relevant in this respect as investing abroad involves setting up or taking over a new plant, therefore causing a higher involvement in the host country. According to this view, if the macroeconomic conditions of the host country do not deteriorate by much, MNEs may show a more rooted character.

On the other side, a distinguished characteristic of MNEs is that, as compared to non-MNEs, they can respond more easily to adverse changes in the macroeconomic variables affecting the host country by simply shifting production to another country. Hence, foreign MNEs can be less rooted in the local economy than local firms as well as domestic MNEs (and non-MNEs). Empirical results trying to account for such effects are quite mixed: with respect to the first hypothesis, Baldwin and Yan (2011)consider the role of real-exchange-rate movements and tariff reductions in affecting the exit rate in Canada, finding that foreign firms and firms that export are more likely to stay in the market than are local firms. Instead, Van Beveren (2007) controls for plant specific differences, showing that domestic and foreign MNEs in Belgium were less rooted in the host country than domestic MNEs and national firms. The second hypothesis is confirmed by looking at the results of Bernard and Sjöholm (2003) and Görg and Strobl (2003a), who use data for Ireland and Indonesia and look at whether there are differences in the probabilities of exit between foreign and domestic firms. They both find that foreign firms are more likely to exit. Non-significant results are found as well: for example, Taymaz and Özler (2007), once they control for industry or plant characteristics, find that in the Turkish case, between 1983 and 2001, differences in terms of survival probabilities disappear. This all points to a high specificity of results. The case of Germany has been analyzed as well: Wagner and Weche Gelübcke (2012) as well as Andrews et al. (2012) have found foreign owned firms to have exit probabilities equivalent to those of the whole population of German owned firms. However, Wagner and Weche Gelübcke (2012) show that foreign owned firms have lower survival probabilities than those of German

firms that are also dependent affiliates and are therefore part of a company network just as are their foreign owned competitors.

Instead, we see that much less attention has been paid to the investigation of the effects that MNEs may have on the survival of domestic firms. Indeed, empirical studies documenting this effect are very scarce. From a theoretical point of view, MNEs may enhance the competition effect in the same industry in which they invest through FDI, pushing some firms out of the market. Two opposing effects can shape the market dynamics: FDI can expand the set of technologies available to local firms but, at the same time, changing the market structure, they enhance competition. The potential disadvantage is that local firms may not be able to adopt the new technologies immediately and, as a consequence, their productivity (at least in the short run) declines, showing negative spillover effects. Usually, the overall effect we observe works in the direction of crowding out local firms, because of the reduction in prices that causes a decrease in the profit function. As a consequence, less efficient firms will be pushed out of the market.

One of the first papers to account for such an effect is Görg and Strobl (2003b) who, using plant-level data for the Irish manufacturing sector in 1973–1996, find that MNEs do not negatively influence the survival of Irish plants. Kosova (2010) analyzes the case of the Czech Republic over 1994–2001, finding that when foreign firms first enter, domestic firms are going to experience several disadvantages, but, after this initial effect has vanished, they may receive substantial benefits from the expanding foreign industry. Instead, Burke et al. (2008), who examine the effects of FDI on the survival of new firms in the UK, find that taking into account the sectoral differences can be very important: they distinguish between static and dynamic industries on the basis of the measure of the firm entry and exit relative to the overall stock of firms. They mainly find a negative effect in dynamic markets while the positive knowledge spillover dominates in the static industries.

Alvarez and Görg (2009) continue in this stream of research, investigating the Chilean case and finding that the presence of MNEs has a positive impact on the survival of local firms: the effect is completely captured by enhancement in TFP. This means that by including TFP in their regressions, the positive effects of FDI disappear. The case of another extra-European country, namely China, is examined by Anwar and Sun (2012): they concentrate their analysis on market entry and exit in period 2003–2007. One important novelty they introduce in their analysis is the disaggregation of FDI according to the country of origin, distinguishing between FDI from Western countries and FDI from the area of Hong Kong, Macau, and Taiwan (HMT). They discover that FDI from HMT has contributed more than FDI from all the other countries to increase the probability of firm exit, while FDI from other countries has made a significant contribution to the rate of firm entry. The reason for the former result could lie in the fact that FDI from HMT is from countries in which the managerial practices are quite similar to those used in China, causing local firms to face higher competition.⁴ Recently, Bandick (2010) examined the case of Sweden, separating the effect caused by foreign MNEs, domestic MNEs, exporting firms, and firms merely domestically oriented. The findings reveal that MNEs negatively affect the survival of domestic firms while a foreign presence does not influence the exit rate of Swedish MNEs. Moreover, they find that a foreign presence has a negative impact specifically on those firms that are non-exporting. With respect to the case of Slovenia, Kejzar (2011) considers what is the role for FDI in the firm selection process in 1994–2003. His results can be mainly summarized by saying that FDI seems to have a negative effect on local incumbent firms but not on those that have a higher export propensity. This effect especially holds when considering the intraindustry spillover effect. This reinforces the idea that the export status of the firm is a relevant variable to take into consideration, as firms belonging to this group benefit from having international exposure in terms of a lower death probability. In the Turkish case, no effect is detected: Taymaz and Ozler (2007) find that, in the long term, foreign presence seems not to affect the probabilities that local firms will

⁴ The ability of MNEs to influence the rate of domestic entry is the focus of a parallel research question: one of the most relevant empirical studies has been carried out in Ayyagari and Kosova (2010), which examines whether FDI can create positive entry spillovers by bringing new technological knowledge, thus causing the entrance of domestic firms into the country. They use data from 1994–2000 for 245 three-digit industries in the Czech Republic, finding that foreign presence can stimulate the entry of domestic firms especially through inter-industry spillovers.

exit the market. Finally, the Italian case is examined by Ferragina et al. (2012), who carry out the analysis for the period from 2004 to 2008 based on a Cox proportional hazard model. Besides finding that foreign MNEs are more likely to exit the market than national firms both in the manufacturing and service sectors, they also find that MNEs contribute to positively affect local firms' life only in the service sectors.

3 Data and variables of main interest

The database used involves mainly three data sources. The first source is the monthly and annual reports of establishments from the manufacturing, mining, and quarrying sectors administered by the German statistical offices. The information is aggregated at the enterprise level and available in the form of annual results for all German firms which employ at least 20 persons and operate in the corresponding industries (for more information see Konold (2007)). These data are of particularly high quality because firms in Germany are legally required to respond to these surveys.

Firm exits are defined as firms which reported to the monthly or annual survey in the year t but not in t + 1. Unfortunately, surviving and exiting firms can only be identified for the year 2007 because in 2009 the monthly reports were adjusted to the German industry classification WZ2008 and thus the population of covered firms was subject to changes. The definition of exits should be treated with caution as there may be identification errors. For example, if a firm relocates abroad or shifts its activities to another sector, outside manufacturing, it is not obliged to report anymore and is regarded as exiting although it survived. However, while this would not be an exit in terms of a firm's death, it is an exit from the particular market or mode of serving it. Another possibility of misidentification would be if firms shrink below the threshold of 20 persons and become exempt from the obligation to report. According to the staff of the Research Data Center, cases of any type of relocation rarely happen and the latter possibility of a firm's shrinking is alleviated due to the fact that most firms go on reporting even if they fall below the threshold simply because they do not realize that they are not obliged to report anymore. In 2007, for manufacturing, 1,802 firm exits were identified, 120 of which were foreign controlled enterprises and are not the focus of this analysis, thus, the exit rate of domestically owned firms is 4.85%.

A second source of information is the Enterprise Group Database created by the German Federal Statistical Office to comply with EU regulation (EC) 716/2007. Since 2007, the European Union legislation demands harmonized statistics on foreign controlled enterprises in each member state (e.g., Vergina and Grell (2009)). A foreign controlled enterprise,⁵ is defined as an enterprise of which more than 50% is owned by a legal or natural person situated abroad. Capital shares as well as voting rights and other forms of control are considered, such as indirect or effective minority control (Eurostat, 2009).⁶ For being able to provide Foreign Affiliates Statistics (FATS) for Germany, the institutions in charge had to purchase information on ownership structures from the commercial data vendor Bureau van Dijk and to integrate this into the national business register (*Unternehmensregister*). Therefore, industry and topic specific surveys recently have become available for analyses related to foreign ownership (for a detailed description of this new database see Weche Gelübcke (2011)).

The role of FDI in a specific industry of the German economy is measured by the role of foreign controlled firms therein. The importance of foreign presence in Germany is not mirrored by the number of foreign owned firms, as they account only for about 1% of all firms in the German non-financial sector (Nahm 2011). Their economic weight is more adequately addressed by considering their employment and turnover shares. In 2007, foreign owned firms employed 13% of the workforce

⁶ Indirect control refers to the an enterprise A's being controlled by enterprise B, both being domestic companies, when enterprise B is, in fact, controlled by an entity abroad. Then, enterprise A is also foreign controlled. Effective minority control is when several minority owners with shares of more than 50% in sum act in concert.

⁵ The terms *foreign controlled*, *foreign owned*, and *foreign* are used interchangeably in this text. The term multinational enterprise (MNE) is also used mainly to describe foreign multinationals as we are not able to identify domestic MNEs. MNEs form the vast majority of foreign controlled firms in Germany although it should be kept in mind that the controlling unit of a foreign enterprise can be, for instance, a natural person.

and generated 23% of the total turnover (ibid.). Consequently, foreign presence is measured as the percentage share of employees working for foreign owned firms in a 2-digit industry. The percentage share of turnover generated by foreign owned firms in certain industries is also considered but reported only for robustness purposes as the results are fairly similar. The descriptive statistics in Table 2 in the following section show that the average foreign employment share across industries is about 18%.

The information as to whether or not a firm imported from other EU or extra-EU countries stems from a third source, namely the German Turnover Tax Statistics Panel (see Vogel and Dittrich (2008) for detailed information).⁷ Since the turnover tax statistics cover firms with a turnover of at least 17,500 EUR, information about importing activities can only be obtained for those. However, this threshold is quite low, considering that only enterprises of more than 20 employees are covered by the monthly and annual surveys.

The data for this study were merged within the AFiD-Project (Official Firm Data for Germany) for the available years 2007 to 2009, and were analyzed via remote access due to the confidentiality of the micro-data (on the AFiD-Project, see Malchin and Voshage (2009)).⁸ For this analysis, the observations were restricted to enterprises from the manufacturing sector in accordance with the NACE classification.

 $^{^{7}}$ Wagner (2011) used this information on imports in the context of firm survival for the first time.

⁸ All computations were programmed in Stata 12 and carried out within the Research Data Center of the statistical office Berlin-Brandenburg.

4 Econometric strategy and covariates

The model through which we analyze a firm's exit decision is a probit model:

$$Pr(Exit_a = 1) = Pr(a > b*), \tag{1}$$

where b^{*} is the cost threshold above which a firm exits from the market. Indeed, at that critical point, a firm earns zero expected profits. Therefore, a firm will exit if its costs are higher than the cut-off point which causes it to earn negative profits. Our main research question is that of investigating whether the cost level of a firm may depend on firm- and industry-level factors, among which the focus of our analysis is the FDI variable.

$$a = x\beta = \alpha + \delta V_{it} + \phi Z_{it} \tag{2}$$

Here, β is a vector of coefficients and X is a vector that includes both firm characteristics and industry level characteristics. We can write the equation as in (3), where Φ is the standard normal cumulative distribution:

$$Pr(Exit_a = 1) = \Phi(x\beta) = \Phi(\alpha + \delta V_{it} + \phi Z_{it})$$
(3)

The firm and industry characteristics introduced are calculated according to the definitions given in Table 1 and are discussed in the following. Some descriptive statistics of these control variables can be found in Table 2.

• *Firm age*: the dummy variable through which we know whether the firm existed before the year 1996 is a proxy for the firm's age. This variable allows us to distinguish older from younger firms: according to the theoretical literature, a firm's exit from the market is a decreasing function of its age. The main reason is that older firms staying in the market for a longer period have learned how to deal with managing day to day businesses. The so called 'liability of newness' is theoretically described by Bellone et al. (2008), who point out how younger

firms, as they may suffer from a lack of reputation and a more difficult access to finance, may be more at risk when competing in imperfectly competitive markets. This effect has been empirically tested also in the case of Germany: Fackler et al. (2012) find that the probability of exit is much higher for younger firms, as the probability of exit for a firm in the first year of life is 12% higher than for firms that are 25 years old. Therefore we expect a negative sign for this variable.

- *Firm Size*: this variable, measured as the number of employees per firm, should be a proxy for the so called effect of 'liability of smallness,' according to which smaller firms usually have lower probabilities of survival. This may be for several reasons: they can be financially constrained and they can suffer from higher costs as they cannot fully exploit economies of scale. Fackler et al. (2012) empirically verify this effect for Germany, confirming that a negative sign is going to be expected.
- *Export Status*: firms operating in foreign markets usually display a higher level of productivity than firms that operate only in the domestic market as they can diversify the risk more easily, substituting domestic sales with foreign sales. Therefore they should have higher probabilities of survival than firms that only produce for the local market, even though it is also likely that firms that export suffer more from sudden shocks of demand that may happen abroad. Moreover, the export status of a firm can also capture a sort of absorption capacity effect. Firms that are more export oriented should also be able to get in touch with knowledge coming from foreign sources. Therefore, we mainly expect a negative sign for this variable.
- *Import Status*: as some recent literature points out, importing firms are more productive than non-importing firms. The reason is that buying cheaper and more technologically advanced intermediate inputs from abroad can cause the firm to be more profitable. Therefore, the probability of exit for these firms is lower as well.

- *Two-way traders*: this type of firm are those that benefit both from exporting and importing activities, besides having usually higher productivity than firms that only import or export. For this reasons, we expect them to have the highest probability of survival.
- Market Concentration: the variable measuring the Herfindahl concentration ratio (HHI) at the industry level (2-digit NACE) should account for a double effect: a higher market concentration can result in a higher price-cost margin which should cause a higher survival probability for firms. Nevertheless, a second effect can occur: firms in highly concentrated markets can suffer from aggressive behavior on the part of competitors, reducing their chances of survival. From an empirical point of view, both positive (e.g., Görg and Strobl (2004)) and negative effects (e.g., Mata and Portugal (2002)) are found. Hence, we do not have a priori expectations for the sign of this variable.
- *Productivity*: this variable is proxied by labor productivity (measured as the amount of sales per employee); it plays a crucial role in the theoretical models by Jovanovic (1982) of industrial dynamics as firms with different levels of productivity have different probabilities of entry, exit, and growth. Therefore we expect that more productive firms are going to have lower exit probabilities.

[Table 1 about here]

[Table 2 about here]

We also include in this specification a set of 3-digit industry dummies in order to control for unobserved heterogeneity across industries.⁹ Our estimation strategy has several steps: in Section 5.1, we first estimate a benchmark model for the full sample of firms in which we take into consideration the role played by the foreign presence variable in enhancing the probability of exit. Then, we split the sample between

⁹ As we estimate the relatively rare event of firm exits, the consideration of 3-digit industry dummies leads to the exclusion of certain industries due to perfect predictions. However, the consideration of differences between industries seems indispensable and outweights the disadvantage of their inclusion.

East and West Germany in order to account for possible differences between the two regions.¹⁰ In Section 5.1 we also split firms by international status (exporters vs. importers). We further explore what are the exit probabilities for several other subsamples, to take into account any heterogeneous impact of foreign presence: to differentiate between high-R&D and low-R&D regions as well as high-tech and low-tech sectors, we ran separate regressions (Section 5.2). Moreover, we consider heterogeneity on the side of the FDI by performing estimates for different categories of FDI: first by dividing them according to their country of origin and then according to the type of investor (Section 5.3). This detailed analysis has never yet been carried out. Lastly, in Section 5.4 we analyze the effect of FDI on the probabilities of local firms' exit from import and export markets.

5 Results

5.1 Exit probabilities and foreign presence

Table 3 gives the estimates of the exit probabilities of domestically owned firms according to the model introduced in the previous section. Starting with benchmark estimates for the full sample, we notice that all controls, except our main variable of interest, foreign presence, have a negative sign. For the dummy variables, the reported marginal effect at the sample mean measures the change in probability of exit when the value of the dummy is 1 instead of 0. As expected, the dummies which account for the international status of the firm have a negative and statistically significant sign. In line with the literature, this stands for the fact that being present in international markets helps the firm to successfully survive the higher competition introduced by foreign MNEs. However, a sort of ranking can be made among the three categories of internationalization: by looking at the size of the marginal effects,

¹⁰ Even more than twenty years after the German reunification in 1990, the two German regions still differ markedly in economic terms. This has been recently confirmed by a 2011 report on East Germany's economic situation and perspectives, carried out by leading German research institutes (IWH et al. (2011)). For instance, the eastern part is much less integrated into the international division of labor, although significant efforts were made by the government to attract FDI.

we note that the highest is the one for two-way traders, followed by the one for importers and then the one for exporters. In this respect, the marginal effect for exporters is not only the lowest but also the level of significance is 5% and not 1%, as for the other coefficients. This means that firms that import or are two-way traders have higher probabilities of survival than firms that only export. This seems to be perfectly in line with what is found by Wagner (2011).

As to these benchmark results, we also notice that, in line with expectations, firm age and size are both negative. Instead, we note that productivity is not significant. The explanation we might give for this unexpected result is that this variable can pick up the productivity spillover effect which is one of the two effects that FDI can have on host country firms. As it is not significant, it means that firms do not absorb any productivity spillover effect and therefore are overwhelmed by the crowding out effect.

The Herfindahl index captures the level of market concentration at the industry level: the sign of the coefficient is negative and it is statistically highly significant. According to our hypotheses, this means that firms in more concentrated markets can achieve a higher level of profitability and therefore remain in the market. It is interesting to note that these control variables maintain the same sign and size of their marginal effects over all regressions for the benchmark model.

As for our main variable of interest, looking at column 1 we first notice that foreign presence is positive and statistically significant, supporting the idea that foreign presence may actually contribute to reduce the probability that a German firm survives. This fact also helps in understanding that, even before the crisis started, the German markets were already suffering from the tougher level of competition introduced by foreigns MNEs. As we are estimating horizontal FDI spillover effects, through this result we also contribute to better understand why inconclusive results are mainly found in empirical studies. By avoiding taking into account the crowding out effect, the literature misses an important part of the story.

As the economic structure of the two main macro-regions, viz., East and West Germany, is still quite relevant, we have taken this fact into account by performing separate regressions for the two subsamples. We still find a negative and statistically highly significant sign of the coefficients although, by looking at the size of the marginal effects, we see that in West Germany the displacement effect is higher.¹¹

Although the marginal effect of foreign presence is statistically significant, this does not necessarily imply its economic relevance. A marginal effect of 0.001 for the West German sample (Table 3) appears fairly small at a first glance. This means that the estimated exit probability of an average indigenous firm increases by about 0.1 percentage point if the foreign presence increases by one percentage point.¹² If we consider the overall exit rate of German firms in our sample, which is 4.85%, an increase by 0.1 seems non-negligible.

As a way to further evaluate the magnitude of the spillover effect, we calculated the marginal effects along the distribution of foreign presence (Table 4). The pattern we note is that the marginal effect happens to be stable in the benchmark regressions, and that it increases in the high- and low-tech sector subsamples with increasing percentile of foreign presence.¹³

[Table 3 about here]

[Table 4 about here]

Moreover, due to the different size of the effect with differing international status of the firm, we decided to dig deeper into the issue by running regressions separately for exporters and importers to explore what is the difference in exit probabilities by trade status¹⁴: the results are almost the same both with respect to sign, statistical significance, and size of the marginal effect. That is, they are both negative, and statistically highly significant. This seems to point to the fact that the international

¹¹ A comparison between different samples has to be regarded with caution here since we look at marginal effects at the sample means and the reported effects refer to different mean values.

¹² Due to the non-linearity of our estimated model, the change in Y may depend on the relative unit size of X. However, in our case, the range of the foreign presence variable seems sufficient for obtaining reliable results.

¹³ Results for the other subsamples are not reported as they do not provide additional information.

¹⁴ Our dataset unfortunately does not allow distinguishing German MNEs from domestic market oriented firms among the exporter and importer samples.

behavior they adopt does not significantly influence their ability to survive competition in the domestic market. To put it another way, even if having relationships with foreign markets is positive for the survival probability, it is not relevant whether the firm carries out importing or exporting activities.

We carried out some robustness checks for these benchmark estimates to see whether the results change on altering the specification of the model or the way the main variable of interest is built. In the first place, we estimated a model in which we did not include the variable measuring productivity, finding robust results because the sign and significance of the coefficients were the same. As this variable should take up the effect of a possible spillover effect, this means that the horizontal spillover effect is not so important for firms, suggesting that it offsets only a minor part of the higher competitive pressure introduced by MNEs.¹⁵ Secondly, we ran regressions building foreign presence using the share of foreign sales in the total sales in each sector. As we can see in Table 5, the sign and significance results are robust, confirming the crowding out effect. However, the size of the coefficient is different in East and West Germany with respect to the FDI measure in terms of employees. This same pattern is observed in Table 6 where we measure FDI using a third methodology, using the ratio of foreign firms to the total firms per sector. Even using this measure, we find that the results are mainly robust although the size of the coefficients are higher both with respect to the overall sample and the split between East and West Germany.

[Table 5 about here]

[Table 6 about here]

5.2 Differences by regional and sectoral R&D intensities

Some papers have investigated the role played by technological factors in explaining the differences in survival probabilities of firms: in this respect, Agarwal and Audretsch (2001) explain how the technological intensity of the industry and the life

¹⁵ Results are not reported but available from the authors upon request.

cycle of the product can affect the relation between firm size and its survival. They find that when the industry has reached a mature stage, smaller firms do not necessarily suffer more from a competition effect. As a matter of fact, one of the industry determinants of firm survival which is seldom considered is the technological regime of which they are part. As our data do not allow us to fully measure the effect that R&D expenditure can exert on the exit probability, we are nevertheless able to account for the technology intensity of the sector to which they belong. For this reason, we ran regressions in Table 7 considering the difference between high-tech and low-tech sectors. We note that high-tech sectors display a negative sign, unlike low-tech sectors. This effect could be for two reasons: the first is that MNEs usually are endowed with a technological knowledge which is higher than that of local firms, and the second one is that firms in low-tech industries do not have enough ability to deal with such advanced technology. Instead, firms in the German high-tech sector can quite easily absorb that technology, contributing to counteract the negative crowding out effect and therefore remain in the market. We repeated the exercise carried out in the previous subsection, to evaluate the pattern of the marginal effect over the foreign presence distribution: the results (Table 4) show how the marginal effects in the high-tech sector always maintain their negative sign throughout the whole distribution despite its decreasing in size for higher percentiles. The reverse happens for low-tech sectors.

The same occurs when we consider differences between R&D intensive regions and low-R&D regions. We divided the sample by using the median value of R&D expenses by administrative region (*Länder*) and considering as high-R&D those regions with a value above the median.¹⁶ We confirm the results found when splitting the sample between high- and low-tech sectors but with a difference: in this case, looking at the size of the marginal effects, the 'surviving' effect in high-R&D regions

¹⁶ To get the data on firms' R&D spending, we used information from the German Cost Structure Survey, which is a representative random sample of around 18,000 firms from manufacturing, mining, and quarrying with at least 20 employees (for more information see Fritsch et al. (2004)). We aggregated the firm level information on in-house R&D expenses at the industry level. The descriptive statistics are in Table 2.

is lower than the crowding out effect in low-R&D regions.¹⁷ Among the possible reasons for such results we can single out that foreign firms operating in those regions for which the sample is quite small can be R&D intensive, therefore inducing a higher technology gap effect. As a consequence, this could negatively influence the survival probabilities of local firms, which will not be able to handle that type of foreign knowledge due to their low absorbing capabilities.

[Table 7 about here]

5.3 Considering heterogeneous FDI

One of the reasons for which studies about the productivity spillover effect from FDI reach inconclusive results may be the fact that heterogeneities on the supply side, that is on the part of the MNEs, are not taken into consideration. Some recent attempts have been made to differentiate technology intensive affiliates (e.g., Marin and Sasidharan (2010), or to take into account the country of origin of the affiliate (e.g., Du et al. (2012)). Introducing this type of heterogeneity can be useful for accounting for the FDI spillover potential which can affect to a great extent the possibility that local firms learn from FDI. If we adopt a more business oriented perspective, by getting in touch with a larger variety of technologies and management practices of MNEs coming from different countries, domestic firms are exposed to a more stimulating environment that can affect the abilities of firms to learn from different sources of knowledge. This element can prove of extreme importance also in the perspective of survival analysis. Ayyagari and Kosova (2010) utilize this level of disaggregation but limiting the analysis to the probability of entry of local firms, finding that the results depend on the country of origin. We carry out the same analysis but for the exit probability.

We measure the foreign presence by country of origin as the share of employees working for firms owned by units located in that country. In particular we look at the categories U.S., European, and other. Descriptive statistics can be found in Table 2, and Table 8 shows the regression results. The first part of Table 8 indicates

 $^{^{17}\,}$ Again it has to be mentioned that we compare marginal effects at different sample means.

that for the full sample, and disentangling between East and West Germany, there is not a substantial difference with respect to sign, significance, or magnitude of the coefficients. This means that merely introducing heterogeneity on the side of the MNEs does not allow us to pick up significantly different effects, that is, the crowding out effect is confirmed.

Instead, when we look at the difference between low- and high-tech industries, we notice that the FDI coming from the U.S. exerts a significant crowding out effect only when considering low-tech industries, but positively contributes to make the life of firms in high-tech sectors longer. Contrary to what is found by Anwar and Sun (2012), we notice that similarities in business and management practices favor the occurrence of a positive knowledge spillover effect, therefore smoothing the competition effect. Instead, U.S. firms can change the market structure towards a stiffer competitive pressure on local firms.

When considering the second type of hetereogneity on the side of local firms, that is, by taking into account the international status of the firm, we see that for importers, the crowding out effect is lower if we consider U.S. firms or other foreign MNEs.

Due to data constraints, we are not able to distinguish affiliates by individual R&D intensities, but we can introduce in our model another type of heterogeneity, that is, the difference between particular investor types. It seems self evident that it makes a difference whether a German manufacturing enterprise is owned by a foreign bank for financial purposes or another industrial company. In the latter case, the investment can be assumed to be part of an operating strategy while in the former it is more likely to be driven by portfolio investment considerations, even if the shares exceed 50%. In order to account for general differences between strategic and financial investments, we look separately at the impact of foreign presence through firms with a financial group head and with an industrial group head.¹⁸ This is a novelty in the literature, as so far this distinction has never been used in empirical

¹⁸ The category 'other' captures group heads such as natural persons, research institutes, states, and governments. FDI by other industrial companies make up by far the major share of the total foreign presence, as can be seen from Table 2.

studies.

We immediately notice a different pattern, that is, in the full sample (and most of the subsamples), financial investments do not contribute to the crowding out effect, unlike industrial FDI, although this seems to be driven by the East German sample. Quite interestingly, we also note that in the high-tech sector, the effect reverses: the reason is that, as shown in the previous subsection, firms in the high-tech sector are more technologically endowed to absorb foreign knowledge and survive. Instead, the negative sign of the coefficient for financial investment could be for two reasons: the first is a low spillover potential. Financial investment, in our analysis, comprises firms which were mainly bought by institutions for portfolio reasons and the group head does not play such an important role in their operations. The assumption of an internal transfer of competitive advantages from financial group heads to producing firms is much more restrictive than in the case of industrial FDI (exceptions may be credit access and managerial expertise). The second, which is nevertheless connected with the first, is that financially owned firms work efficiently anyway regardless of their owner and source their advantages from the German business environment, meaning they are not a 'new' element in the respective market structure and therefore not an element introducing additional competition.

[Table 8 about here]

5.4 Considering export and import market exits

The literature on the relation between trade and survival has not received enough attention so far, even though this research question may prove of extreme importance for matters of policy. Firms that stop importing or exporting usually act in this way mainly because their performance is not sufficient to compete in foreign markets. Indeed, it is well established in the literature that exporting positively contributes to productivity as a sort of learning by exporting effect is detected. It is also true that more productive firms tend to self-select into the export market. The same reasoning can be extended to the importing side: for example, Vogel and Wagner (2010) find that the importing activities of firms exerted a positive impact on their productivity, since importing firms displayed a significant differential in productivity with respect to those firms that did not import. Therefore, we expect that firms ceasing to be exporters or importers may suffer from decreasing productivity. While some of the literature has explored whether exporting or importing activities affect the survival probabilities of firms in local markets (e.g., Wagner (2011); Lopez (2006)), our analytical perspective is a bit different as our research question revolves around the possibility that a firm may stop having an international activity, whether exporting or importing. Indeed, the empirical evidence related to the determinants of the export market dynamics gathered so far on this issue is extremely scarce: Harris and Li (2011) explore what determines the ability of new exporters to survive in export markets in the UK over the period 1997–2003. They mainly find that firm level variables such as size and foreign ownership as well as micro-finance factors (such as profitability) negatively influence the probability of exit. Alvarez and Lopez (2008) examine the same research question but for the probability of entry and exit from both import and export markets for the Chilean case between 1990 and 1999. They find that differences in firms' productivity is one of the main determinants of firm survival in international markets, while trade costs and different factor intensities among industries do not play a crucial role in influencing plant turnover. Nevertheless, the role played by FDI in this process is never examined.

Table 9 shows the results of estimating export and import market exits defined as firms which export/import in 2007 and stopped doing so in the following period. As to the determinants of survival in export markets, firstly, we find that control variables always appear with a negative sign. We notice that productivity both for exports and imports is now statistically significant at the 5% level. This stands for the fact that, as expected, firms with higher productivity will also be those firms that are able to survive in import and export markets. Nevertheless, the effect generated by foreign firms is quite different in the two cases: as to exporting activities, firms are able to survive tougher competition absorbing spillover effects coming from FDI, while the opposite occurs for importing firms, for which the probabilities of exit from the market are higher. This should stand for the fact that exporting firms are positively stimulated by foreign firms in their international activities. This can in some way confirm a sort of export spillover effect for Germany: it means that foreign firms, by raising the productivity of the local firms, can also have beneficial effects on the exporting activities of those firms. Instead, the reason why FDI can lower the probabilities of local firms to survive in import markets may be the fact that firms may source inputs from foreign firms and no longer need to import them. This is only a tentative explanation, as we do not have data on which type of inputs they need and import.

While the result for the exporting firms is confirmed even when we split the sample into West and East Germany, in the import case, the effect turns negative when considering West Germany, therefore indicating a higher probability of survival.

[Table 9 about here]

6 Conclusions

The globalization process has involved a steady increase in the amount of firms that invest abroad through FDI. Despite this historical fact, studies that investigate what effects FDI can have in the dynamics of the local market are relatively scarce. The present paper tries to fill this gap: its research question is what is the effect of the presence of MNEs on the probability of exit of domestic firms of the host country in which they carry out the FDI. This line of research is relevant also from a policy perspective, as the policy instruments to encourage FDI as well as those to prevent the exit of firms from the market can overlap. Indeed, even though most of the literature has concentrated on the firm level and sectoral level determinants of turbulence in the population of firms, the role played by foreign firms has never been examined in the case of Germany. We therefore present the first empirical evidence on this issue, bearing in mind also the limitations of the paper. First and foremost we have to consider that cross-sectional data does not allow us to control for endogeneity.

Our research question can be considered of great interest also in that a deep knowledge of the demography of firms, which possessed high rates of turnover, can highlight the characteristics of the German economy just before the financial crisis started. Our analysis produced some interesting contributions: the first is that we notice a crowding out effect when we take into consideration the full sample. Moreover, when we consider separately East and West Germany, the effect seems to be more pronounced in West Germany. The second interesting result we singled out is relative to the splitting of the sample into high-tech and low-tech sectors and regions: in both cases we can detect a negative effect of FDI on the probability of firm exit only when considering R&D intensive sectors and regions. This suggests the idea that local firms have greater absorptive capacities and therefore are more able to stay in the market by absorbing the spillover effect coming from the foreign presence. Therefore, this effect prevails over the crowding out effect. The third important result achieved is that when considering FDI heterogeneity, we find different results when distinguishing between FDI from different countries or with different types of owners. While in the former case, only European FDI do not display a crowding out effect, that is, when only low-tech industries are considered. Instead, industrial FDI always contributes to crowd out local industries, but not when high-tech sectors are examined. The opposite behavior is exhibited by financial investments.

The industrial organization literature has a long tradition in studying firm turnover issues by mainly referring to the idea that the market structure is one of the best predictors of different entry and exit patterns. However, the same attention has not been paid to the investigation of the determinants of entry and exit into international markets. This is true also for studies considering the effects foreign firms have on the ability of local markets to remain in international markets not only with exporting activities but also with importing activities. Therefore we also investigated how a foreign presence affects the probability of exits from export and import markets: we noticed a positive effect on the likelihood of import market exits, but a negative effect on the likelihood of export market exits. This seems to point in the direction of underlining the differences between firms that carry out exporting and importing activities.

References

- Agarwal, R. and D. Audretsch (2001). Does entry size matter: the impact of technology and product life-cycle on firm survival. *Journal of Industrial Economics* 49, 21–44.
- Aitken, B. and A. Harrison (1999). Do domestic firms benefit from direct foreign investment? Evidence from Venezuela. American Economic Review 89, 605–618.
- Alfaro, L. and M. X. Chen (2012). Surviving the global financial crisis: Foreign ownership and establishment performance. American Economic Journal: Economic Policy 4(3), 30–55.
- Alvarez, A. and R. Lopez (2008). Entry and exit in international markets: Evidence from Chilean data. *Review of International Economics* 16(4), 692–708.
- Alvarez, R. and H. Görg (2009). Multinationals and plant exit: Evidence from Chile. International Review of Economics and Finance 18, 45–51.
- Andrews, M., L. Bellmann, T. Schank, and R. Upward (2012). Foreign-owned plants and job security. *Review of World Economics* 148(1), 89–117.
- Anwar, S. and S. Sun (2012). FDI and market entry/exit. Journal of Asian Economics 23, 487–498.
- Ayyagari, M. and R. Kosova (2010). Does FDI facilitate domestic entry? Evidence from the Czech Republic. *Review of International Economics* 18(1), 14–29.
- Baldwin, J. and B. Yan (2011). The death of Canadian manufacturing plants: heterogeneous responses to changes in tariffs and real exchange rates. *Review of World Economics* 147 (1), 131–167.
- Bandick, R. (2010). Multinationals and plant survival. Review of World Economics 146(4), 609–634.
- Bellone, F., P. Musso, L. Nesta, and M. Quere (2008). Market selection along the firm life cycle. *Industrial and Corporate Change* 17, 753–777.
- Bernard, A. and F. Sjöholm (2003). Foreign owners and plant survival. *NBER* Working Paper 10039.
- Burke, A., H. Görg, and A. Hanley (2008). The impact of foreign direct investment on new firm survival in the UK: evidence for static versus dynamic industries. *Small Business Economics* 31, 395–407.
- Du, L., A. Harrison, and G. Jefferson (2012). Testing for horizontal and vertical foreign investment spillovers in China, 1998-2007. *Journal of Asian Economics* 23(3), 234–243.
- Fackler, D., C. Schnabel, and J. Wagner (2012). Establishment exits in Germany: the role of size and age. *IZA Discussion Paper Series DP No.6349*.

- Ferragina, A., R. Pittiglio, and F. Reganati (2012). Multinational status and firm exit in the Italian manufacturing and service sectors. *Structural Change and Economic Dynamics* 23(4), 363–372.
- Fritsch, M., B. Görzig, O. Hennchen, and A. Stephan (2004). Cost structure surveys for Germany. Schmollers Jahrbuch/Journal of Applied Social Science Studies 124(4), 557–566.
- Garcia, F., B. Jin, and R. Salomon (2012). Does inward foreign direct investment improve the innovative performance of local firms? *Research Policy* 42, 231–244.
- Godard, O., H. Görg, and A. Hanley (2012). Surviving the crisis: foreign multinationals versus domestic firms. *The World Economy*.
- Görg, H. and E. Strobl (2003a). Footloose multinationals. *The Manchester School* 71 (1), 1–19.
- Görg, H. and E. Strobl (2003b). Multinational companies, technology spillovers and plant survival. *Scandinavian Journal of Economics* 105(4), 581–595.
- Görg, H. and E. Strobl (2004). Foreign direct investment and local economic development: beyond productivity spillovers. In: GEP The Leverhulme Centre for Research on Globalisation and Economic Policy. University of Nottingham.
- Harris, R. I. and Q. Li (2011). The determinants of firm exit from exporting: evidence for the UK. International Journal of Economics of Business 18(3), 381–397.
- IWH, DIW, ifo Dresden, IAB, HOF, and RWI (2011). Wirtschaftlicher Stand und Perspektiven f
 ür Ostdeutschland. Studie im Auftrag des Bundesministeriums des Innern. Halle (Saale).
- Javorcik, B. S. (2004). Does Foreign Direct Investment Increase the Productivity of Domestic Firms? In Search of Spillovers through Backward Linkages. American Economic Review 94(3), 605–627.
- Jovanovic, B. (1982). Selection and the evolution of industry. *Econometrica* 50, 649–670.
- Kejzar, K. (2011). The role of foreign direct investment in the host-country firm selection process: firm-level evidence from Slovenian manufacturing. *Review of World Economics* 147, 169–193.
- Kneller, R. and M. Pisu (2007). Industrial linkages and export spillover from FDI. World Economy 30, 105–134.
- Konold, M. (2007). New possibilities for economic research through integration of establishment-level panel data of German official statistics. Schmollers Jahrbuch/Journal of Applied Social Science Studies 127 (2), 321–334.
- Kosova, R. (2010). Do foreign firms crowd out domestic firms? Evidence from the Czech Republic. The Review of Economics and Statistics 92, 861–881.

- Lopez (2006). Imports of intermediate inputs and plant survival. *Economics Let*ters 92(1), 58–62.
- Malchin, A. and R. Voshage (2009). Official firm data for germany. Schmollers Jahrbuch/Journal of Applied Social Science Studies 129(3), 501–513.
- Mansfield, E. (1962). Entry, Gibrat's law, innovation, and the growth of firms. American Economic Review 52, 1023–51.
- Marin, A. and S. Sasidharan (2010). Heterogeneous MNC subsidiaries and technological spillovers: Explaining positive and negative effects in India. *Research Policy* 39, 1227–1241.
- Mata, J. and P. Portugal (2002). The survival of new domestic and foreign-owned firms. *Strategic Management Journal* 23(4), 323–343.
- Nahm, M. (2011). Inward-FATS Auslandskontrollierte Unternehmen in Deutschland 2008. Wirtschaft und Statistik, 899–906.
- Taymaz, E. and S. Ozler (2007). Foreign ownership, competition and survival dynamics. *Review of Industrial Organization* 31, 23–42.
- Van Beveren, I. (2007). Footloose multinationals in Belgium? Review of World Economics 143 (3), 483–507.
- Vergina, G. and M. Grell (2009). Foreign affiliates statistics (FATS). in: Eurostat (ed.). European Economic Statistics, Luxembourg, 107–113.
- Vogel, A. and S. Dittrich (2008). The German turnover tax statistics panel. Schmollers Jahrbuch/Journal of Applied Social Science Studies 128 (4), 661– 670.
- Vogel, A. and J. Wagner (2010). Higher productivity in importing German manufacturing firms: self-selection, learning from importing, or both? *Review of World Economics* 145(4), 641–665.
- Wagner, J. (2011). Exports, imports and firm survival: First evidence from German manufacturing industries. University of Lüneburg, Working Paper Series in Economics.
- Wagner, J. and J. P. Weche Gelübcke (2012). Foreign ownership and firm survival: First evidence for enterprises in Germany. *International Economics* 132(4), 117– 139.
- Weche Gelübcke, J. P. (2011). Ownership patterns and enterprise groups in german structural business statistics. Schmollers Jahrbuch/Journal of Applied Social Science Studies 131(4), 635–647.

Labour productivity	Annual turnover per employee in 1,000 EUR
Firm size	Annual mean over employed persons at the end of each month
Importer status	Indicator variable = 1 if the firm was charged an import turnover tax on inputs from intra-EU or extra-EU countries, 0 otherwise
Exporter status	Indicator variable = 1 if export intensity > 0 , 0 otherwise
Firm age	Indicator variable = 1 if firms was established before 1996, 0 otherwise
$Market\ concentration$	Sum of squared market shares of all firms in a 2-digit industry $(\cdot 10,000)$ (Herfindahl Index)
Turnover	Annual turnover in 1,000 EUR

Table 1: Definition of covariates

Notes: All variables were calculated from the monthly and annual reports of establishments from the manufacturing, mining, and quarrying sectors. Only information on the importer status stems from the German Turnover Tax Statistics Panel.

Variable	Level	M ean	Sd	P10	P25	P50	P75	P90
Foreign presence (employees, $\%$)	industry	17.60	7.23	12.13	12.73	17.74	22.66	26.91
European presence (employees, $\%$)	industry	11.33	5.13	6.06	8.93	10.01	13.02	18.21
U.S. presence (employees, $\%$)	industry	4.21	2.83	0.95	2.11	3.94	5.60	8.52
Industrial presence (employees, $\%$)	industry	14.35	6.87	9.53	9.79	14.09	19.07	23.76
Financial presence (employees, $\%$)	industry	0.98	0.92	0.20	0.39	0.73	1.19	1.81
Market concentration (HHI)	industry	159.34	443.39	12.76	23.08	32.78	63.61	428.4
R&D expenses (million EUR)	industry	2054.78	2906.34	31.22	206.78	678.89	4,538.94	4,923.
Firm size (persons)	firm	155.23	1,239.44	25	34	55.08	114.75	260.5
Labor productivity (1,000 EUR)	firm	179.73	307.51	51.97	83.06	127.35	199.78	320.3
Export intensity $(\%)$	firm	20.11	25.01	0	0	8.33	34.40	60.07
Binary variables								
Variable	Level		7	Number			Share (%,	(
Firm exits	firm			1,802			4.70	
Exporter only	firm			10,979			28.66	
Importer only	firm			3,733			9.74	
Two-way trader	firm			15,498			40.45	
Firm age	firm			19,365			50.54	
High-tech sector	firm			13,803			36.03	

Table 2: Descriptive statistics (2007)

	$Full\ sample$	West Germany	East Germany	Exporters	Importers
Foreign presence (employees, %)	0.0003^{***}	0.001^{***}	0.0004^{***}	0.0003^{***}	0.0003^{***}
	(16.45)	(7.64)	(2.64)	(5.75)	(52.43)
Exporter only (indicator)	-0.007**	-0.006**	-0.007	, I	I
	(2.58)	(2.10)	(0.97)		
Importer only (indicator)	-0.022***	-0.018^{***}	-0.036^{***}	ı	I
	(7.24)	(4.69)	(5.58)		
Two-way trader (indicator)	-0.051^{***}	-0.048***	-0.056***	-0.043^{***}	-0.012^{***}
	(15.64)	(13.83)	(7.75)	(16.55)	(4.20)
Firm size (persons)	-0.00003***	-0.00003***	-0.0001^{***}	-0.00002^{**}	-0.0001^{***}
	(3.04)	(2.70)	(2.96)	(2.49)	(2.88)
Firm age (indicator)	-0.012***	-0.012^{***}	-0.008	-0.014^{***}	-0.003*
	(4.70)	(4.80)	(1.22)	(5.34)	(1.91)
Labor productivity (1,000 EUR)	-1.37e-06	1.63e-07	-0.00001	2.68e-06	-0.0001
	(0.29)	(0.03)	(0.79)	(0.55)	(1.14)
Market concentration -	-6.82e-06***	-8.82e-06***	-0.00003^{***}	-5.21e-06***	-8.72e-06***
	(30.40)	(3.39)	(6.75)	(3.17)	(41.30)
3-digit industry dummies	yes	yes	yes	yes	yes
Observations	34,509	28,396	5,810	23,024	16,734
$McFadden's R^2$	0.057	0.057	0.081	0.066	0.061

Table 3: Exit probability estimates

Sample	Variable	P5	P10	P25	P50	P75	P90	P95
Full sample	foreign presence	0.0003	0.0003	0.0003	0.0003	0.0003	0.0003	0.0003
		(16.77)	(16.10)	(16.05)	(15.59)	(15.18)	(14.84)	(14.57)
High-tech sector	foreign presence	-0.0007	-0.0007	-0.0007	-0.0007	-0.0007	-0.0006	-0.0006
		(6.66)	(6.72)	(6.72)	(6.72)	(7.49)	(8.54)	(8.54)
Low-tech sector	foreign presence	0.0008	0.0009	0.0009	0.0009	0.0011	0.0012	0.0012
		(16.75)	(14.81)	(14.41)	(14.17)	(11.99)	(11.35)	(11.29)
<i>Note:</i> Reported are employees; For the v	marginal effects with variable values at perc	z-values in the set in the set $z = 1$	1 parenthes the descript	is; All forei tive statisti	gn presence cs.	variables n	neasured in	terms of

Table 4: Marginal effects along the distribution of foreign presence

	Full sample	West Germany	East Germany
Foreign presence (sales, %)	0.0003***	0.0003***	0.001***
	(16.45)	(12.17)	(10.14)
Exporter only (indicator)	-0.007**	-0.007**	-0.007
	(2.58)	(2.10)	(0.97)
Importer only (indicator)	-0.022***	-0.018***	-0.036***
	(7.24)	(4.69)	(5.58)
Two-way trader (indicator)	-0.051***	-0.048***	-0.056***
	(15.64)	(13.83)	(7.75)
Firm size (persons)	-0.00003***	-0.00003***	-0.0001***
	(3.04)	(2.70)	(2.96)
Firm age (indicator)	-0.012***	-0.012***	-0.008
	(4.70)	(4.80)	(1.22)
Labor productivity $(1,000 \text{ EUR})$	-1.37e-06	1.63 e-07	-0.0001
	(0.29)	(0.03)	(0.79)
Market concentration (HHI)	-5.90e-06***	$2.48e-06^{***}$	-0.00003***
	(23.03)	(9.74)	(34.17)
3-digit industry dummies	yes	yes	yes
Observations	34,509	28,396	5,810
McFadden's \mathbb{R}^2	0.057	0.057	0.081

Table 5: Exit probabilities (foreign presence measured as sales)

Note: Reported are marginal effects with |z-values| in parenthesis; The market concentration was calculated on the particular regional level for each subsample; Standard errors are adjusted for industry clusters; Significance at the 10% (*), 5% (**) and 1% (***) level.

	Full sample	West Germany	East Germany
Foreign presence (no. of firms, %)	0.001***	0.001***	0.001***
	(16.45)	(7.64)	(2.64)
Exporter only (indicator)	-0.007**	-0.007**	-0.007
	(2.58)	(2.10)	(0.97)
Importer only (indicator)	-0.022***	-0.018***	-0.036***
	(7.24)	(4.69)	(5.58)
Two-way trader (indicator)	-0.051***	-0.048***	-0.056***
	(15.64)	(13.83)	(7.75)
Firm size (persons)	-0.00003***	-0.00003***	-0.0001***
	(3.04)	(2.70)	(2.96)
Firm age (indicator)	-0.012***	-0.012***	-0.008
	(4.70)	(4.80)	(1.22)
Labor productivity $(1,000 \text{ EUR})$	-1.37e-06	1.63e-07	-0.00001
	(0.29)	(0.03)	(0.79)
Market concentration (HHI)	-7.04e-06***	$-8.45e-06^{***}$	-0.00003***
	(32.23)	(3.30)	(7.38)
3-digit industry dummies	yes	yes	yes
Observations	34,509	28,396	5,810
McFadden's \mathbb{R}^2	0.057	0.057	0.081

Table 6: Exit probabilities (foreign presence measured as no. of foreign firms)

Note: Reported are marginal effects with |z-values| in parenthesis; The market concentration was calculated on the particular regional level for each subsample; Standard errors are adjusted for industry clusters; Significance at the 10% (*), 5% (**) and 1% (***) level.

	High-R $&D$ regions	$Low-R { { \it e } { \it O } { \it D } } regions$	High-tech industries	$Low-tech\ industries$
Foreign presence (employees, $\%$)	-0.0003***	0.001^{***}	-0.001***	0.001***
	(6.56)	(12.70)	(6.64)	(13.40)
Exporter only (indicator)	-0.007*	-0.008	0.003	-0.011^{***}
	(1.92)	(1.44)	(0.79)	(3.29)
Importer only (indicator)	-0.019^{***}	-0.028***	-0.012^{*}	-0.025^{***}
	(4.62)	(4.72)	(1.91)	(7.40)
Two-way trader (indicator)	-0.048***	-0.053^{***}	-0.047***	-0.047***
	(12.44)	(9.03)	(9.53)	(11.59)
Firm size (persons)	-0.00003^{**}	-0.0001^{**}	-0.00001^{**}	-0.0001^{***}
	(2.31)	(2.59)	(2.48)	(3.76)
Firm age (indicator)	-0.011^{***}	-0.015^{***}	-0.017***	-0.009**
	(3.71)	(3.45)	(5.66)	(2.58)
Labor productivity (1,000 EUR)	-2.19e-06	2.86e-06	8.42e-06	-8.38e-06
	(0.34)	(0.23)	(0.97)	(1.23)
Market concentration (HHI)	-6.47e-06***	-0.0001^{***}	$8.90e-06^{***}$	-0.00004***
	(15.01)	(22.36)	(29.74)	(13.80)
3-digit industry dummies	yes	yes	yes	yes
Observations	24,690	9,538	11,818	22,691
$McFadden's R^2$	0.058	0.068	0.073	0.054
<i>Note:</i> Reported are marginal effects concentration was calculated on the high-red sectors according to the Ol	with <i>z-values</i> in paren : particular regional lev ECD sectoral annuoach	thesis; Standard errors el for each subsample; · Sionificance at the 10	s are adjusted for industr High-tech industries in 1% (*) 5% (**) and 1%	y clusters; The market clude the medium and (***) level

Table 7: Exit probabilities by regional and sectoral technological intensity

Sample	Z	Europe	U.S.	Other	Industrial	Financial	Other
Full sample	34,509	0.001^{***}	0.001^{***}	0.001^{***}	0.0003^{***}	-0.004***	-0.011^{***}
4		(16.45)	(16.45)	(7.02)	(16.45)	(19.78)	(75.13)
West Germany	28,396	0.001^{***}	0.001^{***}	0.002^{***}	0.0005^{***}	0.002^{***}	-0.005***
2		(22.01)	(7.64)	(7.64)	(7.64)	(6.29)	(6.85)
East Germany	5,810	0.001^{***}	0.001^{***}	0.002^{***}	0.0004^{***}	-0.02***	-0.013^{***}
		(5.01)	(2.99)	(2.64)	(2.64)	(7.00)	(10.27)
High-tech industries	11,818	-0.0002^{**}	-0.002***	-0.004***	-0.001***	0.0003	0.004^{***}
		(2.13)	(6.45)	(6.45)	(6.64)	(0.65)	(6.45)
Low-tech industries	22,691	-0.002***	0.002^{***}	0.003^{***}	0.001^{***}	-0.005***	-0.012^{***}
		(23.69)	(13.40)	(13.40)	(13.40)	(16.81)	(17.14)
Only exporters	23,024	0.001^{***}	0.001^{***}	0.002^{***}	0.0003^{***}	-0.005***	-0.012^{***}
		(5.75)	(35.71)	(5.75)	(5.75)	(4.53)	(81.58)
Only importers	16,734	0.001^{***}	0.0001^{**}	0.0002^{**}	0.0000^{**}	-0.0004	-0.005***
		(52.43)	(2.26)	(2.26)	(2.26)	(0.60)	(82.69)
<i>Note:</i> Reported are ma full set of control variab Significance at the 10%	rrginal effe ales and sta (*), 5% (*	cts of Model andard errors **) and 1% (²	1 estimates s are adjustec ***) level.	with <i>z-value</i> for industry	s in parenthe clusters as in	sis; All mode the previous	ls include a regressions;

Table 8: Exit probability by origin and type of FDI

36

	$Full\ sample$	$West\ Germany$	East Germany
Export market exits	No. of exits $= 770$	No. of exits $= 587$	No. of exits $= 183$
Foreign presence (employees, $\%$)	-0.001***	-0.001***	-0.005***
	(15.28)	(33.62)	(12.38)
Firm size (persons)	-0.0001^{**}	-0.00003^{*}	-0.0001
	(2.20)	(1.89)	(0.84)
Firm age (indicator)	-0.013^{***}	-0.006***	-0.019*
	(5.50)	(3.33)	(1.79)
Labor productivity (1,000 EUR)	-0.00004^{**}	-5.24e-06	-0.0001^{**}
	(2.21)	(0.78)	(2.01)
Market concentration (HHI)	$1.49e-06^{***}$	-3.20e-06***	0.00001^{***}
	(4.14)	(3.44)	(8.25)
Observations	22,408	27,430	2,735
Import market exits	No. of exits $= 1,443$	No. of exits $= 1,171$	No. of exits $= 272$
Foreign presence (employees, $\%$)	0.0004^{***}	-0.009***	0.0002^{**}
	(11.59)	(13.18)	(2.08)
Firm size (persons)	-0.00002	-0.0002	-0.0001
	(1.38)	(1.15)	(1.44)
Firm age (indicator)	-0.030^{***}	-0.031^{***}	-0.031^{**}
	(6.67)	(6.74)	(2.40)
Labor productivity (1,000 EUR)	-0.0001^{**}	-0.0001^{***}	-9.16e-06
	(2.07)	(4.74)	(0.28)
Market concentration (HHI)	-0.00003***	0.0001^{***}	$-3.42e-06^{**}$
	(40.47)	(14.84)	(2.05)
Observations	16,974	13,993	2,555
<i>Note:</i> Reported are marginal effection calculated on the particular regindustry clusters; All specification (**) and 1% (***) level.	ects with $ z-values $ in f on all level for each sult include 3-digit indust	parenthesis; The mark osample; Standard err ry controls; Significanc	et concentration was ors are adjusted for e at the 10% (*), 5%

Table 9: Export and import market exit probabilities

Working Paper Series in Economics

(recent issues)

- No.263: Joachim Wagner: Are low-productive exporters marginal exporters? Evidence from Germany, February 2013
- No.262: Sanne Hiller, Philipp J. H. Schröder, and Allan Sørensen: Export market exit and firm survival: theory and first evidence, January 2013
- No.261: Institut für Volkswirtschaftslehre: Forschungsbericht 2012, Januar 2013
- No.260: Alexander Vogel and Joachim Wagner. The Impact of R&D Activities on Exports of German Business Services Enterprises : First Evidence from a continuous treatment approach, December 2012
- No.259: *Christian Pfeifer*: Base Salaries, Bonus Payments, and Work Absence among Managers in a German Company, December 2012
- No.258: Daniel Fackler, Claus Schnabel, and Joachim Wagner. Lingering illness or sudden death? Pre-exit employment developments in German establishments, December 2012
- No.257: Horst Raff and Joachim Wagner. Productivity and the Product Scope of Multi-product Firms: A Test of Feenstra-Ma, December 2012 [published in: Economics Bulletin, 33 (2013), 1, 415-419]
- No.256: Christian Pfeifer and Joachim Wagner. Is innovative firm behavior correlated with age and gender composition of the workforce? Evidence from a new type of data for German enterprises, December 2012
- No.255: *Maximilian Benner*. Cluster Policy as a Development Strategy. Case Studies from the Middle East and North Africa, December 2012
- No.254: Joachim Wagner und John P. Weche Gelübcke: Firmendatenbasiertes Benchmarking der Industrie und des Dienstleistungssektors in Niedersachsen – Methodisches Konzept und Anwendungen (Projektbericht), Dezember 2012
- No.253: Joachim Wagner: The Great Export Recovery in German Manufacturing Industries, 2009/2010, November 2012
- No.252: Joachim Wagner. Daten des IAB-Betriebspanels und Firmenpaneldaten aus Erhebungen der Amtlichen Statistik – substitutive oder komplementäre Inputs für die Empirische Wirtschaftsforschung?, Oktober 2012
- No.251: Joachim Wagner: Credit constraints and exports: Evidence for German manufacturing enterprises, October 2012
- No.250: Joachim Wagner: Productivity and the extensive margins of trade in German manufacturing firms: Evidence from a non-parametric test, September 2012 [published in: Economics Bulletin 32 (2012), 4, 3061-3070]
- No.249: John P. Weche Gelübcke: Foreign and Domestic Takeovers in Germany: First Comparative Evidence on the Post-acquisition Target Performance using new Data, September 2012
- No.248: Roland Olbrich, Martin Quaas, and Stefan Baumgärtner. Characterizing commercial cattle farms in Namibia: risk, management and sustainability, August 2012
- No.247: Alexander Vogel and Joachim Wagner: Exports, R&D and Productivity in German Business Services Firms: A test of the Bustos-model, August 2012
- No.246: Alexander Vogel and Joachim Wagner. Innovations and Exports of German Business Services Enterprises: First evidence from a new type of firm data, August 2012

- No.245: Stephan Humpert: Somewhere over the Rainbow: Sexual Orientation Discrimination in Germany, July 2012
- No.244: *Joachim Wagner*: Exports, R&D and Productivity: A test of the Bustos-model with German enterprise data, June 2012 [published in: Economics Bulletin, 32 (2012), 3, 1942-1948]
- No.243: Joachim Wagner: Trading many goods with many countries: Exporters and importers from German manufacturing industries, June 2012 [published in: Jahrbuch für Wirtschaftswissenschaften/Review of Economics, 63 (2012), 2, 170-186]
- No.242: *Joachim Wagner*. German multiple-product, multiple-destination exporters: Bernard-Redding-Schott under test, June 2012 [published in: Economics Bulletin, 32 (2012), 2, 1708-1714]
- No.241: Joachim Fünfgelt and Stefan Baumgärtner. Regulation of morally responsible agents with motivation crowding, June 2012
- No.240: John P. Weche Gelübcke: Foreign and Domestic Takeovers: Cherry-picking and Lemon-grabbing, April 2012
- No.239: *Markus Leibrecht* and *Aleksandra Riedl*: Modelling FDI based on a spatially augmented gravity model: Evidence for Central and Eastern European Countries, April 2012
- No.238: Norbert Olah, Thomas Huth und Dirk Löhr: Monetarismus mit Liquiditätsprämie Von Friedmans optimaler Inflationsrate zur optimalen Liquidität, April 2012
- No.237: *Markus Leibrecht* and *Johann Scharler*. Government Size and Business Cycle Volatility; How Important Are Credit Contraints?, April 2012
- No.236: *Frank Schmielewski* and *Thomas Wein*: Are private banks the better banks? An insight into the principal-agent structure and risk-taking behavior of German banks, April 2012
- No.235: Stephan Humpert: Age and Gender Differences in Job Opportunities, March 2012
- No.234: Joachim Fünfgelt and Stefan Baumgärtner. A utilitarian notion of responsibility for sustainability, March 2012
- No.233: *Joachim Wagner*: The Microstructure of the Great Export Collapse in German Manufacturing Industries, 2008/2009, February 2012 [published in: Economics - The Open-Access, Open-Assessment E-Journal, Vol. 7, 2013-5]
- No.232: Christian Pfeifer and Joachim Wagner: Age and gender composition of the workforce, productivity and profits: Evidence from a new type of data for German enterprises, February 2012
- No.231: Daniel Fackler, Claus Schnabel, and Joachim Wagner. Establishment exits in Germany: the role of size and age, February 2012
- No.230: Institut für Volkswirtschaftslehre: Forschungsbericht 2011, January 2012
- No.229: *Frank Schmielewski*: Leveraging and risk taking within the German banking system: Evidence from the financial crisis in 2007 and 2008, January 2012
- No.228: Daniel Schmidt and Frank Schmielewski: Consumer reaction on tumbling funds Evidence from retail fund outflows during the financial crisis 2007/2008, January 2012
- No.227: Joachim Wagner. New Methods for the Analysis of Links between International Firm Activities and Firm Performance: A Practitioner's Guide, January 2012

- No.226: Alexander Vogel and Joachim Wagner: The Quality of the KombiFiD-Sample of Business Services Enterprises: Evidence from a Replication Study, January 2012 [published in: Schmollers Jahrbuch/Journal of Applied Social Science Studies 132 (2012), 3, 379-392]
- No.225: Stefanie Glotzbach: Environmental justice in agricultural systems. An evaluation of success factors and barriers by the example of the Philippine farmer network MASIPAG, January 2012
- No.224: Joachim Wagner: Average wage, qualification of the workforce and export performance in German enterprises: Evidence from KombiFiD data, January 2012 [published in: Journal for Labour Market Research, 45 (2012), 2, 161-170]
- No.223: *Maria Olivares* and *Heike Wetzel*: Competing in the Higher Education Market: Empirical Evidence for Economies of Scale and Scope in German Higher Education Institutions, December 2011
- No.222: Maximilian Benner: How export-led growth can lead to take-off, December 2011
- No.221: Joachim Wagner and John P. Weche Gelübcke: Foreign Ownership and Firm Survival: First evidence for enterprises in Germany, December 2011 [published in: International Economics/Économie Internationale, 132(2012), 4, 117-139]
- No.220: *Martin F. Quaas, Daan van Soest,* and *Stefan Baumgärtner*. Complementarity, impatience, and the resilience of natural-resource-dependent economies, November 2011
- No.219: *Joachim Wagner*: The German Manufacturing Sector is a Granular Economy, November 2011 [published in: Applied Economics Letters, 19(2012), 17, 1663-1665]
- No.218: Stefan Baumgärtner, Stefanie Glotzbach, Nikolai Hoberg, Martin F. Quaas, and Klara Stumpf: Trade-offs between justices, economics, and efficiency, November 2011
- No.217: Joachim Wagner. The Quality of the KombiFiD-Sample of Enterprises from Manufacturing Industries: Evidence from a Replication Study, November 2011 [published in: Schmollers Jahrbuch/Journal of Applied Social Science Studies 132 (2012), 3, 393-403]
- No.216: John P. Weche Gelübcke: The Performance of Foreign Affiliates in German Manufacturing: Evidence from a new Database, November 2011 [published in: Review of World Economics, 149(1) (2013), 151-182]
- No.215: Joachim Wagner. Exports, Foreign Direct Investments and Productivity: Are services firms different?, September 2011
- No.214: Stephan Humpert and Christian Pfeifer. Explaining Age and Gender Differences in Employment Rates: A Labor Supply Side Perspective, August 2011
- No.213: John P. Weche Gelübcke: Foreign Ownership and Firm Performance in German Services: First Evidence based on Official Statistics, August 2011 [forthcoming in: The Service Industries Journal]
- No.212: John P. Weche Gelübcke: Ownership Patterns and Enterprise Groups in German Structural Business Statistics, August 2011 [published in: Schmollers Jahrbuch / Journal of Applied Social Science Studies, 131(2011), 4, 635-647]
- No.211: Joachim Wagner: Exports, Imports and Firm Survival: First Evidence for manufacturing enterprises in Germany, August 2011 [published in: Review of World Economics, 149(1) (2013), 113-130]

(see www.leuphana.de/institute/ivwl/publikationen/working-papers.html for a complete list)

Leuphana Universität Lüneburg Institut für Volkswirtschaftslehre Postfach 2440 D-21314 Lüneburg Tel.: ++49 4131 677 2321 email: brodt@leuphana.de

www.leuphana.de/institute/ivwl/publikationen/working-papers.html